Title: The Fitness for the Ageing Brain Study II (FABS II): protocol for a randomized controlled clinical trial evaluating the effect of physical activity on cognitive function in patients with Alzheimer’s disease

Authors:

Elizabeth V Cyarto (e.cyarto@nari.unimelb.edu.au)
Kay L Cox (kay.cox@uwa.edu.au)
Osvaldo P Almeida (osvaldo.almeida@uwa.edu.au)
Leon Flicker (leon.flicker@uwa.edu.au)
David Ames (dames@unimelb.edu.au)
Gerard Byrne (gerard.byrne@uq.edu.au)
Keith D Hill (keith.hill@nh.org.au)
Christopher D Beer (christopher.beer@uwa.edu.au)
Dina LoGiudice (dina.logiudice@mh.org.au)
Kana Appadurai (kana_appadurai@health.qld.gov.au)
Muireann Irish (m.irish@neura.com.au)
Emma Renehan (e.renehan@nari.unimelb.edu.au)
Nicola T Lautenschlager (nicolatl@unimelb.edu.au)

Version: 2 Date: 27 October 2010

Author's response to reviews: see over
October 26, 2010

Trials Editorial
BioMed Central
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HL

Dear Drs Altman, Furberg, Grimshaw and Rothwell,

It is with great pleasure that we submit our manuscript ‘The Fitness for the Ageing Brain Study II (FABS II): protocol for a randomized controlled clinical trial evaluating the effect of physical activity on cognitive function in patients with Alzheimer’s disease’.

We believe our paper should be published in Trials as it concerns a National Health Priority, it is of high scientific quality, and the study presents a novel approach to the management of mild to moderate Alzheimer’s disease.

This manuscript is in alignment with the journal’s editorial policies and we have no competing interests to declare.

We look forward to hearing from you.

Kind regards,

Elizabeth V Cyarto, PhD

(On behalf of the FABS II research team)